Author's response to reviews

Title: Unemployment and ill health: a connection through inflammation?

Authors:

Jukka Hintikka (jukka.hintikka@kuh.fi)
Soili M Lehto (soili.lehto@kuh.fi)
Leo Niskanen (leo.niskanen@kuh.fi)
Anne Huotari (anne.huotari@uku.fi)
Karl-Heinz Herzig (karl-heinz.herzig@oulu.fi)
Heli Koivumaa-Honkanen (heli.koivumaa@kuh.fi)
Kirsi Honkalampi (kirsi.honkalampi@kuh.fi)
Sanna Sinikallio (sanna.sinikallio@kuh.fi)
Heimo Viinamäki (heimo.viinamaki@kuh.fi)

Version: 4 Date: 26 October 2009

Author's response to reviews:

The following revisions have been made:

Major revisions:
1. Acknowledgements - we have added a new Acknowledgements section at the end of the manuscript text, directly preceding the Reference list.
2. We have added the titles of the cited Websites.
3. Reference 10, 16 - The doi numbers of these references have been removed and instead the volume and page numbers (ref. 16) or "in press" (ref. 10) have been added.

Minor revisions:
4. Underlining - we have removed all underlinings from the manuscript
5 Vertical lines within tables - we have removed all vertical lines within tables.